# **ECONSTOR** Make Your Publications Visible.

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Graziano, Clara; Luporini, Annalisa

## Working Paper Optimal delegation when the large shareholder has multiple tasks

CESifo Working Paper, No. 3028

**Provided in Cooperation with:** Ifo Institute – Leibniz Institute for Economic Research at the University of Munich

*Suggested Citation:* Graziano, Clara; Luporini, Annalisa (2010) : Optimal delegation when the large shareholder has multiple tasks, CESifo Working Paper, No. 3028, Center for Economic Studies and ifo Institute (CESifo), Munich

This Version is available at: https://hdl.handle.net/10419/38883

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



## WWW.ECONSTOR.EU

## Optimal Delegation when the Large Shareholder has Multiple Tasks

## Clara Graziano Annalisa Luporini

## CESIFO WORKING PAPER NO. 3028 CATEGORY 11: INDUSTRIAL ORGANISATION APRIL 2010

An electronic version of the paper may be downloaded • from the SSRN website: www.SSRN.com • from the RePEc website: www.RePEc.org • from the CESifo website: www.CESifo-group.org/wp

## Optimal Delegation when the Large Shareholder has Multiple Tasks

## Abstract

We analyze the optimal ownership, delegation and compensation structures when a manager is hired to run a firm and to gather information on investment projects. The initial owner has two tasks: monitoring the manager and supervising project choice. Optimality would require a large ownership stake for monitoring but a small stake for not interfering with managerial incentives. Delegating project choice to the manager can alleviate this conflict if managerial private benefits are not too small. The large shareholder retains full ownership of the firm but monitoring, and the resulting firing policy, are distorted. Severance pay plays a key role in the optimal compensation scheme. Delegation is interpreted as a dual-board structure.

JEL-Code: G34, L22.

Keywords: large shareholder, delegation, monitoring, board of directors, corporate governance.

Clara Graziano University of Udine Department of Economics via Tomadini 30 Italy – 33100 Udine clara.graziano@uniud.it Annalisa Luporini University of Florence Department of Economics via delle Pandette 9 Italy – 50127 Florence luporini@unifi.it

April, 2010

We would like to thank seminar audiences at the University of Padua, Bologna, Leeds, Trento, and Bruno Parigi for helpful comments and suggestions. Any remaining errors are our sole responsibility.

### 1 Introduction

The board of directors performs several tasks. The board oversees the hiring and firing of top executives and in general it has a disciplinary role. At the same time, it also has an advising role and helps to set the strategy of the firm. Each role, in turn, is a combination of several tasks. For example Adams, Hermalin and Weisbach (2008) point out that the assessment of top executives has two components: monitoring what top management does and determining the intrinsic ability of top management. The advising role instead involves communication and information transmission between management and directors in order to select the best strategy.

It is well known that in a multi-task environment, the incentive to induce an agent to perform one task may conflict with the incentive to perform another task. A possible solution to incentive conflicts is represented by an appropriate allocation of tasks among the agents. Aghion and Tirole (1997) have shown that a principal may profitably delegate formal or real authority to an agent in order to mitigate his own incentive to interfere with the agent's decision because such interference would reduce the agent's incentives.

In the present paper we study a problem of allocation of control rights over tasks between the initial owner of a firm and the management. The initial owner has to perform two tasks: elaborating the strategy of the company (project selection) and monitoring/firing the manager. The incentives required by the two tasks conflict each other because "too high" a level of monitoring discourages the manager from exerting effort to gather information needed for project choice. We suggest that delegating the elaboration of the company's strategy to the management may be optimal because it alleviates this incentive conflict. In our setting, the initial owner decides the optimal ownership structure and hires a manager to run the firm and to gather information on investment projects. At the same time the owner decides whether to retain the authority to choose the investment project or to delegate it to the manager. Owner's and manager's preferences over projects are divergent because of managerial private benefits. The owner also has to monitor the manager to find out his ability in order to decide whether to fire or retain him. Profits depend on both tasks and optimality would require different ownership stakes: a large stake is required for monitoring but a small stake is necessary for not interfering with the proposed strategy. We show that by delegating authority over project choice and by using an optimal compensation scheme, the large shareholder is able to retain full ownership of the firm (high incentive for monitoring) and, at the same time, to encourage information gathering from the manager.

When project choice is delegated to the manager, full ownership of the firm is optimally retained by the initial owner but monitoring and the resulting firing decision are somewhat distorted. Monitoring intensity in fact is highest when the initial owner retains the authority to select the project. This happens because monitoring has a twofold effect. A high monitoring intensity increases the probability of replacing a low-ability manager, thus increasing profits. However, by increasing the firing probability, it also reduces managerial effort in information gathering. As a consequence, the initial owner reduces monitoring in order to provide the proper incentive to the manager and this effect is more important when the manager has the crucial task of elaborating the company's strategy.

Managerial discretion in project selection (i.e., delegation) and monetary compensation (incentive pay and severance pay) coexist in the optimal contract and complement each other in providing incentives to the manager. Severance pay is an important component of the contract because of the conflict between the firing policy (dependent on monitoring) and the need to provide incentives to the manager. Severance pay alleviates this conflict by (partly) insulating the manager from the negative consequences of being fired without entrenching him. A noteworthy feature of the optimal contract is that, under delegation, severance pay is higher than incentive pay. In other words, when fired a (lowability) manager receives a higher payment than when he is confirmed because he must be compensated for the loss of private benefits. Delegation in fact allows the manager to choose his preferred project and hence to enjoy private benefits when he is retained by the firm. We interpret the delegation of authority over elaborating the company's strategy as the choice of a dual-board structure where the management board (in charge of project selection), is separated from the supervisory board controlled by the large shareholder (in charge of monitoring and firing/retention decision). Contrary to what happens in public companies with dispersed ownership, in companies with concentrated ownership there often is an 'excessive' involvement of owners in the management of the firm rather than lack of monitoring. Owner's activism is common in Continental Europe and more generally in countries with concentrated ownership (see, e.g. Faccio and Lang, 2002). Indeed, recent empirical studies on corporate governance have suggested that the presence of an active large shareholder is much more common than previously thought (see, among others, La Porta et al., 1999, and Holderness, 2009). The policy implication of our model is that a dual-board structure can be a useful device to alleviate the problem of excessive interference by the large shareholder without reducing the fraction of retained shares.

Our framework is similar to that of Burkart, Gromb and Panunzi (1997) who show that a large shareholder actively involved in a firm's management may reduce managerial discretion and prevent the manager from appropriating private benefits. However, large shareholder activism interferes with the need to motivate the manager to take initiative and make uncontractible investments. The authors suggest that the ownership structure can be used to solve this conflict: a reduction in the large shareholder's stake can limit her incentive to exercise control rights. In their model the large shareholder has only one task (interfering with project choice)<sup>1</sup>. Then, one instrument (the ownership structure) is sufficient to alleviate the incentive conflict. In our model instead, the large shareholder has two tasks (interfering with project choice and monitoring the manager). As a consequence, more instruments must be used to provide

<sup>&</sup>lt;sup>1</sup>Burkart, Gromb and Panunzi, call 'monitoring' such activity of project choice supervision, because in their model there is no role for managerial quality. In our model, on the contrary, we have both a problem of opposite preferences over projects and a problem of discovering the unknown ability of the CEO. Thus, in line with previous models of board behavior (see for example Hermalin and Weisbach (1998), Hirshleifer and Thakor (1998)) we use the word monitoring to indicate the activity of the large shareholder aimed at discovering the ability of the manager in order to decide whether to retain or to fire him. See also Tirole (2001).

proper incentives. Furthermore, we show that using a plurality of instruments allows the initial owner to retain full ownership of the firm. Indeed, if incentive compensation cannot be used for some reason, delegation requires the initial owner to sell a positive fraction of shares, as shown in extension A.

Our paper is also related to the scant literature on multitasking in boards. Most of the fast-growing literature on boards has analyzed various aspects of a single task, trying to assess if the board accomplishes it effectively. For example, several papers study optimal board composition from the shareholders' point of view (see Harris and Raviv 2008, Raheja 2005, Warther 1998 among others). Other papers instead examine the determinants of CEO and directors appointment (Hermalin and Weisbach 1998 and 2003, Hermalin 2005). In general, little attention has been paid to tasks interaction. The few exceptions we are aware of are Adams and Ferreira (2008), Dominguez-Martinez, Swank and Visser (2008) and Graziano and Luporini (2003). Adams and Ferreira study the trade-off between the advisory and the decision making roles of the board. They show that these two functions may conflict: the CEO may refrain from sharing his information with the board because he fears that this information may be used to interfere in decision making. Thus, it may be desirable to separate these functions in two different boards. Dominguez-Martinez, Swank and Visser consider the twin tasks of disciplining and screening the CEO and they show that a tension between these two functions arises when the CEO cares about benefits from holding office. They examine the trade-off between discouraging executives from becoming too active (empire building) and maximizing the probability that the executive is competent. As a result the link between bad performance and dismissal is weakened. Graziano and Luporini focus on the same conflict between CEO selection and CEO dismissal in a setting where the dismissal/retention decision provides information both on the quality of the CEO and on the ability of the board. Since firing the CEO indicates that his selection was wrong in the first place, the board may try to hide its inability in the selection of the CEO by distorting the second task (CEO retention/dismissal decision).

The rest of the paper is organized as follows. Section 2 presents the basic model. Section 3 anticipates some basic features of managerial compensation and delegation. The choice of monitoring intensity by the large shareholder is analyzed in Section 4, while Section 5 illustrates the choice of effort by the manager. Section 6 examines the choice of the optimal contract and of the ownership structure by the large shareholder. Section 7 presents three extensions of the basic model and finally Section 8 discusses the results and illustrates how delegation can be interpreted in terms of a dual board structure.

#### 2 The model

Consider a firm initially owned by a single owner (she) who needs to hire a manager (he) of unknown ability. The manager is needed both to run routine activities and to collect information on prospective new projects. Depending on the delegation structure, such information can be used either by the owner or by the manager himself to select a project. Project choice cannot be verified by third parties; thus the owner cannot contract directly on such choice. There are two types of manager: high-ability (H) and low-ability (L). The ability of the manager depends on the quality of the matching between his skills and the skills required by that particular firm. As a consequence, the manager himself is not aware of his type when joining the firm, and no screening is possible. Only after the manager has spent some time on the job, can the owner assess his ability through monitoring and decide whether to fire or retain him. Both the owner and the manager are risk-neutral. The latter has a reservation level of utility equal to zero, has no wealth and is protected by limited liability.

Our model develops over four periods. In period 1, the owner decides the fraction  $\alpha$  of shares she wants to retain,  $(1 \ge \alpha > 0)$  and makes a take-it-orleave-it contract offer to a manager, randomly selected from a given pool where the fraction of high-ability managers is  $\lambda$ , with  $0 < \lambda < 1$ . The remaining shares, if any, are sold to dispersed shareholders who do not play any active role in firm management. We then refer to the initial owner as the 'large shareholder'. The contract  $C = \{d, w, s, P^F(\hat{t})\}$  offered to the manager specifies the delegation structure d, the compensation w, the severance pay s, and the firing rule  $P^F(\hat{t})$ where  $\hat{t} = \hat{H}, \hat{L}$ , is the manager type reported by the large shareholder.  $d \in$  $\{0, 1\}$  where d = 0 indicates that the large shareholder retains the authority to choose the investment project and d = 1 indicates that project choice is delegated to the manager.<sup>2</sup> For simplicity, the firing rule is expressed in terms of the firing probability  $P^F(\hat{t})$  because, as it will be shown later, the optimal values of  $P^F(\hat{t})$  are either 0 or 1. Given that the firing/retention decision is observable and verifiable, it can be part of the contract.

In period 2, the manager can exert effort in order to obtain information about projects, and, based on such information, the firm (either the large shareholder or the manager himself, depending on the chosen value of d) may decide to undertake a risky project in addition to 'business as usual'. The firm's returns from business as usual are verifiable and depend on manager's type. Returns are  $\overline{\pi} > 0$  if the manager is high-ability, and are  $\pi = 0$  if the manager is lowability. Recalling that the manager is randomly chosen and that  $\lambda$  is the fraction of high-ability managers, the expected return from routine activity is  $\lambda \overline{\pi}$ . As to investment projects, we assume that the firm can select one among  $N \ge 3$ possible projects. However, only two projects are relevant: project 0 and project 1. These projects offer a positive and verifiable monetary return R > 0 with probabilities  $p_0$  and  $p_1$  respectively, where  $p_0 > p_1 > 0$ . With complementary probabilities the return is zero. To simplify matters we assume that project return is independent of manager type. The remaining N-2 projects (indexed from 2 to N-1) yield very negative returns with some probability so that randomly picking a project results in a negative expected return. Project 1 offers private benefits b > 0 to the manager implementing it. For technical reasons we assume that  $b < \lambda \overline{\pi}(1-\lambda) \equiv \beta$ , that  $\beta(2+\lambda) < p_1 R < 2-\beta$  and that  $(p_0 - p_1)R < \beta$ .<sup>3</sup>

<sup>&</sup>lt;sup>2</sup>We could allow for d to be stochastic  $(0 \le d \le 1)$  but, as the large shareholder's problem would not be concave in d, the optimal solutions would still be corner ones. Thus, we simplify the exposition by directly presenting the two cases of d = 0 and d = 1.

<sup>&</sup>lt;sup>3</sup>These assumptions ensure interior solutions for effort, ownership structure, incentive pay and monitoring thus simplifying the exposition. They also ensure that incentive compatibility conditions on s and w are satisfied. Results would not be affected by also considering corner

In order to identify project 0 and project 1, information must be gathered by the manager. Gathering information requires unobservable effort  $e \in [0, 1]$ at cost  $\frac{e^2}{2}$ . A level of effort e implies that the manager is able to identify project 0 and 1 (and to distinguish between them) with probability e. With probability 1 - e, the manager does not obtain enough information and, given that random choice is unprofitable, no project is chosen. Thus e is the probability of undertaking a risky project. If the manager is in charge of project choice, he will use such information to choose his preferred project. If the large shareholder retains control over project choice, the manager will have to pass the information to her. We assume that the information is 'hard' meaning that the manager can either withhold information or pass it on truthfully. Furthermore, we assume that the large shareholder is able to correctly process the information obtained from the manager.<sup>4</sup>

In period 3, once effort has been exerted and (possibly) a risky project has been selected, the large shareholder can engage in monitoring. Monitoring is aimed at assessing the manager's type in order to decide whether to retain or to fire him. Recall that manager's ability positively affects the returns from routine activity so that replacing a low-ability manager with a high-ability one increases such returns. However monitoring is costly and its outcome is uncertain. A monitoring intensity  $M \in [0,1]$  costs  $M^2/2$  to the large shareholder and allows her to learn the manager's ability with probability M. With probability 1 - Mthe large shareholder remains uninformed. Monitoring is uncontractible, so that there is no possibility to induce the large shareholder to increase monitoring intensity through payments contingent on it. Following monitoring, the large shareholder makes a report  $\hat{t}$  on the manager's type. Then, on the basis of this report and of the firing rule  $P^F(\hat{t})$  specified in the contract, the manager is either retained or fired. We assume that the managers' pool is large enough for the probability of a high ability replacement to remain  $\lambda$ . We also assume that there are no firing costs other than the severance pay and that, in the event of

solutions or binding constraints.

 $<sup>^{4}</sup>$  We assume that processing information as well as being in charge of project choice imposes no costs on the large shareholder. Including such a cost would insert an additional motive for delegation.

firing, the new manager cannot modify the investment project.

In period 4, cash flows are realized and payments are made. If project 1 has been implemented and the incumbent manager has not been fired, the latter obtains private benefits b. In section 7 we extend the analysis to the case where private benefits are obtained before the possible firing decision, so that the incumbent manager always obtains b when project 1 is implemented.

Summarizing, timing is as follows:

period 1: The initial owner determines the optimal ownership structure,  $\alpha$  and offers a contract  $C = \{d, w, s, P^F(\hat{t})\}$  to the manager.

period 2: The manager implements effort e and possibly a new investment project is chosen (by either the manager or the large shareholder).

period 3: The large shareholder exerts monitoring and learns manager's type with probability M. Then, she decides whether to retain or fire the manager.

*period* 4: Cash flows from routine activity and possibly from a risky project are obtained, together with private benefits if project 1 is selected. Payments are made.

We proceed by backward induction, finding first the optimal level of monitoring. However, since managerial compensation and delegation structure are determined only in Section 6, we anticipate in the next section their main characteristics that are crucial in understanding the following analysis.

## 3 Managerial Compensation and Delegation

In order to elicit the optimal level of effort from the manager, the large shareholder offers him an incentive contract. Manager's compensation can be conditioned on verifiable returns and on the firing decision, so that the manager may receive severance pay if fired. Given that the manager has no possibility to improve his (exogenously given) ability, and consequently cannot increase returns from routine activity, there is no point in conditioning managerial compensation on  $\overline{\pi}$ . Performance pay, and possibly severance pay, conditional on return R can instead be used to induce a greater effort from the manager. We then consider compensations formed by a base salary, performance pay and severance pay. Given that the reservation utility of the manager is normalized to zero and that routine activity imposes no disutility on the manager, the base salary is equal to zero. Moreover, since the manager is protected by limited liability, we restrict our attention to compensation contracts that yield zero payment when the project's return is zero and a non-negative bonus when the project succeeds. The bonus is w if the manager is retained, and s if he is fired. In other words, also severance pay is contingent on project succeeds. Recalling that in the event of firing the new manager cannot modify the investment project, no incentive compensation is paid to the replacement.

Incentive compensation however is not sufficient to solve the conflict of interests between the manager and the large shareholder. Recall that project 1 offers private benefits b > 0 to the manager implementing it. This, together with the assumption that  $p_0 > p_1$ , creates a conflict between the manager and the large shareholder because, everything else being equal, the former prefers project 1 and the latter, who is interested in monetary return, prefers project 0. Given that project choice in uncontractible, compensation cannot be contingent on project choice. As a consequence the delegation structure d is key to determining which project will be chosen. In case of delegation (d = 1), the manager will choose project 1, and in case of no delegation (d = 0) the large shareholder will choose project 0. Delegation can then be viewed as a commitment device for the large shareholder to implement project 1 in an incomplete contract environment.

### 4 Period 3: Monitoring and firing decision

In period 3, once the manager has exerted effort and a project has possibly been chosen, the large shareholder may find it optimal to monitor the manager in order to learn his type. Depending on the reported type and on the firing rule contained in the contract, the incumbent manager is then retained or replaced.

In the event of firing, the ability of the replacement is unknown at the time he is hired. If a low-ability manager is replaced, there is an increase in expected returns from routine activity equal to  $\lambda \overline{\pi}$ . If instead a high-ability manager is replaced, there is an expected loss equal to  $(1 - \lambda)\overline{\pi}$ . The optimal contract must then prescribe firing (retention) if low (high) ability is observed, i.e.  $P^F(\widehat{L}) = 1$  and  $P^F(\widehat{H}) = 0$ . However, the decision of the large shareholder also depends on the difference between incentive pay w and severance pay s. In order to guarantee that the large shareholder has no incentive to misreport the manager's type given  $P^F(\widehat{H}) = 0$  and  $P^F(\widehat{L}) = 1$ , w and s must satisfy the following two incentive compatibility conditions:

$$(1-\lambda)\overline{\pi} - p_i(w-s) \ge 0 \tag{1}$$

$$\lambda \overline{\pi} - p_i(s - w) \ge 0 \tag{2}$$

with i = 0, 1. The first condition guarantees that the gain in expected profit from retaining a high-ability manager,  $(1-\lambda)\overline{\pi}$ , is not offset by the difference between expected bonus if he is retained and severance pay if he is fired,  $p_i(w-s)$ . Similarly, the second condition guarantees that the gain in firing the low-ability type,  $\lambda\overline{\pi}$ , is not offset by the difference  $p_i(s-w)$ .

Consider now the optimal firing probability when monitoring is unsuccessful. If nothing is learnt from monitoring (which happens with probability 1 - M), there is no expected gain from replacing the manager because the expected profit from the incumbent is equal to the expected profit from a replacement. As a consequence the corresponding incentive compatible firing probability,  $P^F(\hat{0}) \equiv P^F$ , will depend on the sign of w - s:

$$P^{F} \begin{cases} = 1 & \text{if } w > s \\ = 0 & \text{if } w < s \\ \in [0,1] & \text{if } w = s \end{cases}$$
(3)

The optimal values of w, s and  $P^F$  will be determined in Section 6. For the moment suffice it to note that the values of w, s and  $P^F$  contained in the contract satisfy conditions (1), (2) and (3) respectively. Recalling that project 1 (project 0) is chosen under delegation (no delegation), the large shareholder chooses M to maximize her expected profits for d = 0, 1:

$$\max_{M} \quad \alpha_{d} \{ M [\lambda \overline{\pi} + (1 - \lambda)\lambda \overline{\pi}] + (1 - M)\lambda \overline{\pi} + I_{pro} p_{d} [R - w_{d}(M\lambda + (1 - P_{d}^{F})(1 - M)) - s_{d}(M(1 - \lambda) + (1 - M)P_{d}^{F}] \} - \frac{M^{2}}{2}$$

where  $I_{pro} \in \{0, 1\}$  is an indicator function taking value 1 if a risky project has been selected and value 0 otherwise. >From the first-order condition we obtain:

$$M_d = \alpha_d \left\{ (1 - \lambda) \left[ \lambda \overline{\pi} + I_{pro} p_d (1 - \lambda - P_d^F) (w_d - s_d) \right] \right\}.$$
 (4)

Monitoring increases in the expected gain from replacing a bad with a good manager,  $\lambda \overline{\pi}$ , and in the fraction of shares owned by the large shareholder,  $\alpha$ . Given that  $P^F$  must satisfy (3), the level of monitoring is the highest when a risky project is not undertaken ( $I_{pro} = 0$ ) or if a project is implemented with w = s. In such cases, monitoring intensity simplifies to:

$$M_d = \alpha_d (1 - \lambda) \lambda \overline{\pi} \tag{5}$$

This high-lights one aspect of the conflict between monitoring and incentives for project choice. Whenever incentives for effort require to set  $w \neq s$ , the level of monitoring must be reduced with respect to (5) to preserve incentive compatibility in the firing decision.

## 5 Period 2: Managerial effort

In period 2, the manager decides the level of effort that maximizes his expected utility. If the large shareholder retains control rights on investment decision (d = 0), the manager knows that project 0 will be chosen and that he will only get monetary compensation. Under delegation (d = 1) instead he will be able to choose project 1 obtaining also private benefits b.

Recall that the manager does not know his own type when deciding the level of e. However, he can anticipate the value of  $M_d$  that will be chosen by the large shareholder in order to calculate the *ex ante* firing probability in case a risky project is implemented,

$$F_d \equiv M_d(1-\lambda) + (1-M_d)P_d^F = P_d^F + M_d(1-\lambda - P_d^F), \qquad d = 0, 1.$$
(6)

The manager then solves:

$$\max_{e} e\{w_d p_d(1-F_d) + s_d p_d F_d + db(1-F_1)\} - \frac{e^2}{2} \qquad d = 0, 1$$

>From the first order condition we obtain :

$$e_0 = p_0 \left[ w_0 (1 - F_0) + s_0 F_0 \right] \tag{7}$$

and

$$e_1 = p_1 \left[ w_1(1 - F_1) + s_1 F_1 \right] + b(1 - F_1) \tag{8}$$

Managerial effort is an increasing function of the expected value of monetary compensation and, in the case of delegation, of private benefits. As far as monetary compensation is concerned, the expectation is taken with respect to the probability of success conditional on project choice. For a given value of d, when the project succeeds, the bonus is w if the manager is confirmed (which happens with probability 1 - F). If the manager is fired (which happens with probability F) he gets severance pay s. Private benefits instead are obtained only under delegation and if the manager remains with the firm.

Monetary compensation and private benefits are substitutes with respect to managerial effort. As a consequence the large shareholder will choose the combination of w, s, and d that makes it cheaper to motivate the manager.

## 6 Period 1: Large Shareholder maximization problem

The objective of the large shareholder in period 1 is to maximize the equity value of the firm net of monitoring costs, i.e. the sum that she could raise by selling all the shares to dispersed shareholders. The large shareholder decides the optimal ownership structure  $\alpha$  and the contract  $C = \langle d, w, s, P_F(\hat{t}) \rangle$  she

offers to the manager, anticipating how the subsequent choice of effort by the manager,  $e_d$ , and her own choice of monitoring intensity  $M_d$  are influenced by these variables.

>From Section 4 we know that  $P^F(\hat{H}) = 0$  and  $P^F(\hat{L}) = 1$  independently of the delegation structure, so that we are left with the determination of  $\alpha$ , w, s,  $P^F$  and d. Since d is a discrete variable that can take only two values we solve the maximization problem of the large shareholder with respect to  $\alpha$ ,  $P^F$ , w and s for each value of d. Then, by comparing firm equity value with and without delegation, we determine the optimal d.

Using (4) for the value of  $M_d$ , the large shareholder solves the following maximization problem for d = 0, 1:

$$\underset{\alpha,w,s,P^F}{Max} \qquad e_d p_d [R - w \left(1 - F_d\right) - sF_d] + \lambda \overline{\pi} + \tag{9}$$

$$e_d\{\lambda(1-\lambda)\overline{\pi}\alpha[(1-\lambda)\left[\lambda\overline{\pi}+p_d(1-\lambda-P^F)(w-s)\right]] - (10)$$

$$\frac{\left\{\alpha\left[\left(1-\lambda\right)\left[\lambda\pi+p_d\left(1-\lambda-P^2\right)\left[w-s\right)\right]\right\}^2}{2}\right\}+$$
(11)

$$(1 - e_d)[\alpha(1 - \lambda)^2 \lambda^2 \overline{\pi}^2 - \frac{[\alpha(1 - \lambda)\lambda\overline{\pi}]^2}{2}]$$
(12)

where  $e_0$  is determined in (7),  $e_1$  in (8), w - s and  $P^F$  must satisfy (1), (2), (3).

The solution of the problem is characterized in the following proposition.

Proposition 1. When the large shareholder retains control over project choice (d = 0), she optimally sets  $w_0 = s_0 = \frac{R}{2}$  and  $P_0^F \in [0, 1]$ . When she decides to delegate project choice to the manager (d = 1), she optimally sets  $w_1 = \frac{R}{2} - \frac{b+(1-\lambda^2)\lambda\pi}{2p_1} + \frac{b^2(1-\lambda)(1+3\lambda)}{4p_1}$ ,  $s_1 = w_1 + \frac{b}{(1-\lambda)p_1}$  and  $P_1^F = 0$ . Both under d = 0 and d = 1 she retains full ownership of the firm  $\alpha_0 = \alpha_1 = 1$ 

#### Proof. See Appendix.

When the large shareholder finds it optimal to keep control over project choice (d = 0), returns from the project are equally shared between the manager and the large shareholder by setting the bonus equal to  $\frac{R}{2}$  irrespective of the firing decision. As a consequence, the firing probability becomes irrelevant,

 $P_0^F \in [0, 1]$ , and we can assume without loss of generality that the firing rule inserted in the contract prescribes either  $P_0^F = 0$  or  $P_0^F = 1$ . When the large shareholder delegates project choice to the manager (d = 1), monetary compensation is lower because the manager has positive expected private benefits. Note that in this case the level of the severance payment exceeds the level of the incentive pay:  $s_1 > w_1$ . Now  $P_1^F$  must be equal to 0 in order for the contract to be incentive compatible. If  $P_1^F$  were positive, the large shareholder would in fact have an incentive to report high ability even when she has no information. Both under delegation and no delegation, the large shareholder retains full ownership of the firm, which enables her to maximize monitoring intensity. However the realized level of monitoring depends on the delegation structure. By using Proposition 1 and substituting the values of w, s,  $P^F$  and  $\alpha$  in expression (4) we obtain:

$$M_0 = (1 - \lambda)\lambda\overline{\pi}$$
  
 $M_1 = (1 - \lambda)[\lambda\overline{\pi} - b]$ 

which, once substituted into (6), result in a lower level of the *ex ante* firing probability under delegation than under the authority of the large shareholder

$$F_0 = M_0(1-\lambda) + (1-M_0)P_0^F \ge (1-\lambda)M_0 > (1-\lambda)M_1 = F_1$$

Monitoring intensity is highest when the large shareholder retains authority over project choice (d = 0).<sup>5</sup> When instead project choice is delegated to the manager (d = 1), monitoring is reduced to decrease the *ex ante* firing probability  $F_1$ , and to increase the expected value of private benefits,  $(1 - F_1)b$ . The *ex ante* firing probability is optimally reduced to motivate the manager to exert effort since private benefits represent an incentive component obtainable only if the manager stays with the firm. In other words, when d = 1 there is a trade-off between routine profits that are increasing in monitoring and profits from the risky project that require a low monitoring. When d = 0 instead, there is no point in distorting monitoring because monetary compensation is independent

<sup>&</sup>lt;sup>5</sup>In this case the *ex ante* firing probability depends on the chosen value of  $P_0^F$ . Given  $w_0 = s_0$  when monitoring is unsuccessful the large shareholder is indifferent between retaining or firing the incumbent manager.

of the firing policy:  $w_0 = s_0$  and the manager receives no private benefits. These results are summarized in the following proposition.

Proposition 2. When project choice is delegated to the manager, monitoring and the resulting ex ante firing probability are optimally reduced to provide the manager with the proper incentive to exert effort.

It is worth noting the role of severance pay in this incentive conflict. Severance pay, by providing a payment  $s_1 > w_1$ , insulates the manager from the replacement policy, compensating him for the loss of private benefits. In fact when private benefits are unobtainable (d = 0), incentive pay and severance pay are equal. A severance payment higher than incentive pay introduces a distortion in the level of monitoring, and consequently in the firing probability which is reduced with respect to the case of no delegation. Notice that in the absence of severance pay, the only means to motivate the manager would be to reduce the firing probability. Other papers have shown that some degree of entrenchment may be optimal because the firing threat may lead to suboptimal managerial behavior.<sup>6</sup> In our paper, severance pay is important for the opposite reason: it allows the large shareholder to motivate the manager without distorting the firing policy too much. To a certain extent, in our model severance pay is a substitute for a more accommodating firing policy.

High severance pay contracted at the time the manager is hired and contingent on project return that realizes after the manager's departure may look unrealistic. However, these characteristics of the optimal severance payment are consistent with the empirical evidence provided by Yermack (2006) and by Rusticus (2006). Rusticus analyzes a sample of newly hired CEOs from S&P 1500 and finds that half of the CEOs in his sample have some form of severance agreements when they take office. Yermack, in a study of separation pay for retired and dismissed CEO, confirms that severance agreements contracted at the time the manager is hired are common and shows that, in the event of firing, it is a common practice to grant the CEO a discretionary separation pay in ad-

 $<sup>^{6}</sup>$  For example Almazan and Suarez (2003) have demonstrated that shareholders may gain by relinquishing some power in favour of the CEO. In their model, severance pay is an important instrument to this end because it reduces the probability of firing the manager.

dition to the contracted separation pay. Discretionary separation pay includes several items: cash payments and new awards of stock or options, consulting and non-compete agreements that will be paid over a number of years, adjustments to benefit pensions and adjustment to equity compensation. All these discretionary components affect CEOs wealth for several years after CEO departure and it seems reasonable to consider that, for example, the payments received as a consultant or the non-compete agreements are contingent on project success.

Having characterized the optimal contract with and without delegation, we now have to determine if and when delegating project choice is profitable. First of all, by substituting condition (22) from the proof of Proposition 1 and  $\alpha = 1$ in the net equity value of the firm (see the objective of the large shareholder) we obtain the following lemma.

Lemma 1. The net equity values of the firm without and with delegation can be expressed as:

$$V_{0} = (e_{0})^{2} + \lambda \overline{\pi} + \frac{(1-\lambda)^{2}\lambda^{2}\overline{\pi}^{2}}{2}$$

$$V_{1} = (e_{1})^{2} + \lambda \overline{\pi} + \frac{(1-\lambda)^{2}\lambda^{2}\overline{\pi}^{2}}{2}$$
where  $e_{0} = p_{0}\frac{R}{2}$  and  $e_{1} = p_{1}\frac{R}{2} + \frac{b}{2}[1 - (1-\lambda)^{2}(\lambda \overline{\pi} - b)] - \frac{(1-\lambda)^{2}b^{2}}{4}.$ 

This tells us that delegation is optimal if and only if the effort level obtained when d = 1 is higher than the effort level obtained when d = 0. By comparing  $e_0$  with  $e_1$  we can then establish conditions for delegation to be profitable.

Proposition 3. There exists a threshold value  $\overline{b}$  such that for  $b > \overline{b}$  the large shareholder prefers to delegate project choice (d = 1) and for  $b < \overline{b}$  she prefers to keep control over it (d = 0).  $\overline{b}$  is increasing in R and in the difference  $p_0-p_1$ so that high values of both make delegation less likely.

#### Proof: see Appendix.

Proposition 3 establishes a positive relationship between delegation and the level of private benefits. For sufficiently high level of b, project choice is delegated to the manager. In this case (see Proposition 1), monetary compensation is reduced in proportion to the level of expected private benefits. For low levels of b instead, the large shareholder finds it more profitable to keep control over project choice because the reduction in the expected return from the project under delegation (recall that the manager chooses project 1) is not compensated by the reduction in monetary compensation. In other words, the higher is b, the more profitable it is to substitute private benefits for monetary compensation in order to motivate the manager.<sup>7</sup> Given that  $p_0 > p_1$ , the threshold value  $\overline{b}$ is increasing in R because so is the difference in expected returns from project 0 and 1. Obviously such difference is amplified by an increase in the difference  $p_0 - p_1$ , for any given R.

### 7 Extensions

In this section we present three extensions of the basic model. In the first one, we further explore the relationship between delegation, ownership and compensation policy by studying the limit case with no monetary compensation, i.e. with w = s = 0. In the second one, we examine a different timing that allows the manager to enjoy private benefits before the firing/retention decision. This modified timing eliminates the negative impact of monitoring on managerial effort. The third extension maintains this modified timing and exacerbates the conflict of interests between manager and large shareholder by allowing project 0 to yield private benefits to the large shareholder in addition to the higher return. This of course makes delegation less likely.

#### A. No monetary incentives

There may be situations in which monetary incentives cannot be paid because, for example, monetary return R is not verifiable. In the basic model, monetary compensation and delegation are alternative methods of providing incentives. If we rule out monetary incentives, delegation is the only way to induce managerial effort and this obviously reinforces the optimality of delegation. It immediately

<sup>&</sup>lt;sup>7</sup>Notice that for b = 0 the large shareholder is indifferent between d = 0 and d = 1 because the conflict of interests disappears. Even the manager now prefers to choose project 0 that has a higher probability of success.

follows from (7) that in the absence of monetary incentives, managerial effort is equal to zero when d = 0, and is equal to

$$e_{1|w=s=0} = b(1 - F_{1|w=s=0}) \tag{13}$$

when d = 1. Private benefits become now necessary for the manager to exert effort and consequently for a risky project to be chosen. Then the following proposition holds.<sup>8</sup>

Proposition 4. If no monetary compensation is available, i.e. if w = s = 0, the threshold value is  $\overline{b}_{w=s=0} = 0$ , so that the large shareholder always chooses to delegate the manager (d = 1) when private benefits are strictly positive. The fraction of retained shares is strictly smaller than one,  $(\alpha_{1|w=s=0} = 1 - \frac{bRp_1}{\lambda \overline{\pi}})$ and the firing probability when no information is revealed by monitoring is zero  $(P_{1|w=s=0}^F = 0)$ .

#### Proof. See Appendix.

Whenever b > 0 the large shareholder chooses delegation in order to have a positive probability of undertaking the risky project. The fraction of retained shares  $\alpha_{1|w=s=0}$  is now lower than 1. The reason is that, once we exclude monetary compensation, the only way to reduce  $M_1$  is to decrease the value of  $\alpha$ . If  $\alpha_1$  were equal to 1, monitoring intensity, and consequently the *ex ante* firing probability  $F_1$ , would be too high from an incentive point of view because a high level of  $F_1$  implies a relatively low level of effort (see (13)).

If b = 0, the large shareholder is indifferent as to the delegation structure because no risky project will be undertaken. In this case, she also retain full ownership of the firm in order to maximize monitoring and consequently profits from routine activity.

Given that  $\alpha_{1|w=s=0} < 1$ , a time consistency problem may arise. Once the manager has exerted effort and information on investment projects has been gathered, the large shareholder may have an incentive to buy back some shares thus increasing her stake. This would prompt an increase in monitoring intensity

<sup>&</sup>lt;sup>8</sup>In the absence of monetary compensation, the condition  $1 > Rp_1$  must be added to the other assumptions in order to ensure a positive level of  $\alpha$ .

that would increase expected returns from routine activity without having negative effects on the (sunk) level of effort. The manager however, would anticipate the behavior of the large shareholder and would lower the effort level accordingly. In other words, the optimal contract illustrated in the above proposition would be destroyed. Fortunately it can be proved that the large shareholder has no incentive to retrade after date 1, provided that her stake is publicly observable.<sup>9</sup> In order to induce dispersed shareholders to sell their shares, the large shareholder has to offer a price that fully reflects the increase in the equity value of the firm. But if this is the case, she cannot gain by purchasing back the shares.

#### B. Manager's private benefits independent of monitoring

There may be cases where the manager has no need to stay for a long time with the firm in order to enjoy private benefits. For example, the benefits may consist of the increased visibility obtained by the announcement of the new project undertaken by the firm. The case where the firing/retention decision is taken after the manager has enjoyed private benefits can be represented by a change in the timing considered so far, such that in period 3 private benefits (if any) from the risky project are realized; in period 4 the large shareholder implements monitoring and decides whether to confirm or fire the manager; and finally in period 5, routine and project's returns are realized.

When the large shareholder retains control over choice and selects project 0, no private benefits are generated. As a consequence, for d = 0, monitoring, effort and the contract offered by the large shareholder are the same as those analyzed in the basic model of Section 6. Under delegation, instead, effort choice is modified because b is now obtained with certainty when project 1 is selected. It can be easily verified that effort becomes

$$\widetilde{e}_1 = p_1[\widetilde{w}_1(1 - F_1) + \widetilde{s}_1F_1] + b.$$

Then, the following proposition holds.

 $<sup>^{9}</sup>$ The proof is omitted because it parallels the argument provided in Proposition 4 by Burkart, Gromb and Panunzi (1997) to prove that the large shareholder does not have an incentive to renegatiate her effort.

Proposition 5. If the manager is able to enjoy private benefits before the firing/retention decision, the large shareholder optimally sets  $\tilde{w}_0 = \tilde{s}_0 = R/2$  when she retains control over project choice (d = 0), and  $\tilde{w}_1 = \tilde{s}_1 = \frac{R}{2} - \frac{b}{2p_1}$  when she delegates project choice (d = 1). The optimal value of  $\alpha$  is independent of d and is always equal to 1 and  $\tilde{P}^F \in [0, 1]$ .

Proof: See Appendix.

If the manager is able to enjoy private benefits before the firing/retention decision, the incentive provided by delegation is stronger than in the basic model and this makes it cheaper to induce him to exert effort when d = 1. In this case, incentive pay is lower than in the basic model,  $\widetilde{w}_1 < w_1$ . Moreover,  $\widetilde{s}_1 = \widetilde{w}_1$ because, in the event of firing, there is no need to compensate the manager for the loss of private benefits. Equality between incentive and severance pay implies that monitoring is independent of delegation. It then reaches the same level as in the basic model without delegation:  $\widetilde{M}_1 = \widetilde{M}_0 = M_0$ . This in turn means that, under delegation, monitoring intensity is now higher:  $M_1 > M_1$ . In the basic model, the firing decision (which positively depends on monitoring intensity) affects managerial choice of effort because it reduces the probability of enjoying private benefits. To contrast this effect, the large shareholder optimally chooses to increase severance pay above the level of  $w_1$  in order to reduce monitoring intensity and the *ex ante* firing probability. When the link between managerial incentive and monitoring is receded as in the present case, there is no need to reduce monitoring and the bonus paid if the project is successful does not depend on the firing decision, i.e.  $\tilde{s}_1 = \tilde{w}_1$ . This result is summarized in the following proposition.

Proposition 6. If the manager is able to enjoy private benefits before the firing/retention decision, delegation implies no tension between monitoring and managerial incentive to exert effort.

Considering that  $\tilde{e}_1 = p_1 R + \frac{b}{2}$ , we immediately see that effort under delegation is now higher than in the basic model. This implies that delegation becomes optimal for lower values of b as shown by the following proposition. Proposition 7. If private benefits are enjoyable before the firing/retention decision, there exists a threshold value  $\tilde{b}$  such that for  $b > \tilde{b}$  the large shareholder prefers to delegate project choice (d = 1) and for  $b < \tilde{b}$  she prefers to keep control over it (d = 0).  $\tilde{b}$  is increasing in R, p<sub>0</sub>, and decreasing in p<sub>1</sub>. Moreover  $\tilde{b} < \bar{b}$ .

Proof: See Appendix.

#### C. Large Shareholder Private Benefits

Suppose now that incongruence over projects is exacerbated by the presence of private benefits for the large shareholder associated with project 0, where B denotes these benefits. The expressions for monitoring and managerial effort are not affected by large shareholder's private benefits. Also the maximization problem under delegation is the same as in subsection B above. However, for d = 0 the large shareholder's problem becomes:

$$\begin{aligned}
& \underset{\alpha,w,s,P^{F}}{\text{Max}} \qquad \widehat{e}_{0}[(R-w(1-\widehat{F}_{0})-s\widehat{F}_{0})p_{0}+B] + \lambda\overline{\pi} + \\
& \widehat{e}_{0}\{\lambda(1-\lambda)\overline{\pi}\alpha[(1-\lambda)\left[\lambda\overline{\pi}+p_{0}(1-\lambda-P^{F})(w-s)\right]\right] - \\
& \left\{\alpha[(1-\lambda)\left[\lambda\overline{\pi}+p_{0}(1-\lambda-P^{F})(w-s)\right]\right]^{2}
\end{aligned}$$
(14)

$$\frac{\left\{\alpha\left[\left(1-\lambda\right)\left[\lambda\pi+p_{0}\left(1-\lambda-P^{2}\right)\left(w-s\right)\right]\right\}^{2}\right\}}{2}\right\}+$$
(15)

$$(1-e_0)\left[\alpha(1-\lambda)^2\lambda^2\overline{\pi}^2 - \frac{\left[\alpha(1-\lambda)\lambda\overline{\pi}\right]^2}{2}\right]$$
(16)

The solutions for  $\alpha$  and w, s, and  $P^F$  lead to the following results.<sup>10</sup>

Proposition 8. If private benefits for the large shareholder are associated with project 0 and the manager is able to enjoy private benefits before the firing/retention decision, the optimal value of  $\alpha$  is independent of d and always equal to 1 and  $\widehat{P}^F \in [0, 1]$ . Incentive compensation is equal to  $\widehat{w}_0 = \widehat{s}_0 = \frac{R}{2} + \frac{B}{2p_0}$ when d = 0 and to  $\widehat{w}_1 = \widehat{s}_1 = \frac{R}{2} - \frac{b}{2p_1} = \widetilde{w}_1$  when d = 1.

#### Proof: See Appendix.

When d = 0 the monetary compensation paid to the agent is now higher because the payoff that the large shareholder obtains from managerial effort is higher. Under delegation instead (d = 1) the solutions for w, s and  $\alpha$  are the same as in the absence of B. However, the expected loss from implementing

<sup>&</sup>lt;sup>10</sup>In order to ensure an interior solution for  $e_0$  we now assume  $p_0R + B < 2$ .

project 1 instead of project 0 is now higher and consequently, as shown in the next proposition, the large shareholder is more reluctant to delegate project choice. The next proposition shows that we may never have delegation if B is large relatively to b.

Proposition 9. The large shareholder delegates project choice if and only managerial private benefits b are sufficiently higher than her own private benefits B i.e. iff  $b - B > (p_0 - p_1) R$ .

Proof: see Appendix.

Delegation is now chosen only when managerial private benefits are higher than the sum of the large shareholder's private benefits plus the expected loss in returns from choosing project 1 instead of project 0.

### 8 Discussion

We have analyzed the optimal delegation, ownership structure, and compensation scheme in a setting where a large shareholder has two tasks: monitoring the manager to assess his ability and supervising the choice of a risky project. The effort of the manager is needed to gather information on projects but preferences over projects are dissonant because of managerial private benefits. An incentive conflict for the large shareholder then arises because a large ownership stake is optimal for the first task (monitoring) while a small stake is called for the second task (supervising project choice without reducing managerial incentive). The key insight of the paper is that such conflict can be solved by delegating project choice to the manager. This in turn allows the large shareholder to retain full ownership of the firm. However, full ownership comes at the price of distorting monitoring and the resulting firing policy. Delegation is optimal when private benefits are relatively large because the prospect of enjoying such benefits motivates the manager.

Ownership structure, delegation and monitoring are determined together taking into account their interdependence. Monitoring (which is an increasing function of the large shareholder's fraction of shares) positively affects the ex ante firing probability because it helps identifying a low-ability manager. However, a higher firing probability reduces managerial incentives provided by private benefits. Then, under delegation, the optimal incentive scheme prescribes a lower level of monitoring and of the firing probability. This is accomplished by increasing severance pay with respect to incentive pay. Thus, our model contributes to the literature that shows that some managerial entrenchment may be optimal from the shareholders' point of view (see Almazan and Suarez (2003) and Inderst and Muller (2008) ) and offers a rationale for a potentially high severance pay even when the manager is low-ability<sup>11</sup>. Indeed, high severance pay for the low-ability manager results from optimal contracting and not from rent extraction.

A few testable implications can be derived from our basic model. The model predicts that when the manager has discretionary power (delegation) we should observe a lower turnover, longer tenure, higher severance pay and lower incentive pay than when he has no such a power. Furthermore, the severance package should apply only to CEO dismissals and not to voluntary retirements. Finally, the model predicts that delegation (i.e. manager discretionary power) should be more likely the smaller the difference in projects' expected return and the more profitable it is to motivate the manager with private benefits rather than with monetary compensation.

A number of extensions are also analyzed. First, if no monetary incentive is available, delegation is always optimal because private benefits are the only means to motivate the manager. Given that severance pay can no longer be used to reduce the large shareholder's incentive for monitoring, now the fraction of shares retained by the large shareholder is smaller than one. Second, we analyze a different timing in which private benefits are obtained independently of monitoring (i.e. before the possible firing decision). In this case monitoring has no negative effects on incentives, its intensity is higher than in the basic model and this makes delegation optimal in a larger number of cases. Finally, we consider

<sup>&</sup>lt;sup>11</sup>Also Manso (2008) stresses the importance of rewarding failures to incentive the manager.

large shareholder's private benefits assuming that these are associated with the most profitable among the risky projects. In this case, delegation implies a higher forgone return for the large shareholder and consequently becomes less likely.

Provided that delegation is optimal in a number of cases, the question arises of how the large shareholder can credibly commit to delegate project choice. In other words the problem is how to make the choice of d = 1 (when specified in the contract), enforceable. We believe that an appropriate dual board structure may be an effective commitment to delegate project choice to the management. Consider a two-tier structure composed by a supervisory board and a management board with the following features. The positions on the two boards are mutually exclusive so that the same person cannot sit on both boards. The management board is mainly composed of managers with executive functions whose objectives do not conflict with the objective of the CEO.<sup>12</sup> The large shareholder instead sits on the supervisory board that is mainly composed of members whose objectives are aligned with hers. The supervisory board is in charge of monitoring and of making the firing/retention decision while the management board is in charge of project selection. The choice of such a dual board structure then allows the large shareholder not to interfere with managerial decision without losing control of the firing/retention decision.

This interpretation leads to interesting policy implications for those areas, such as Continental Europe, where concentrated ownership is still the norm. Faccio and Lang (2002), for example, report that family ownership is predominant in 11 European countries: Austria, Belgium, Finland, France, Germany, Italy, Norway, Portugal, Spain, Sweden and Switzerland. In some of these countries, i.e. Germany, Austria, Belgium, the dual structure is mandatory, in other countries like France and Italy, companies can choose between different board models. Our result does not suggest that existing two-tier boards represent the

<sup>&</sup>lt;sup>12</sup>This is usually the case for inside members whose career and salary depend on CEO's decisions. Moreover, executive directors may also share some of the CEO's private benefits. For example, if the CEO expands the firm beyond its optimal size for personal prestige and power motives, all members of the management board benefit from the increased visibility of a larger firm. In other words, the preferences of the executive directors are usually aligned with those of the CEO.

optimal corporate governance solution but it indicates that a specific two-tier structure may be the optimal choice for companies under family control. Although limiting the activism of large shareholders can be a challenging task, it is more difficult for large shareholders to overrule or to interfere with management board decisions when the functions of the two boards are separated and clearly defined by corporate charters. Thus, we believe that a two-tier board structure may be a valuable option in Continental Europe as well as in other areas where ownership of firms (including that of large corporations) is concentrated and founding families are often 'too active' in firm management.

### **9** References

Adams, R. and D. Ferreira (2007) "A Theory of Friendly Boards" *Journal of Finance*, 62, 217-250.

Adams, R. Hermalin B.E. and M. Weisbach (2008) "The Role of Boards of Directors in Corporate Governance: A Conceptual Framework and Survey" NBER working paper 14486 (forthcoming in Journal of Economic Literature 2009)

Aghion, P. and J. Tirole (1997) 'Formal and Real Authority in Organizations' Journal of Political Economy, 97, 1-29.

Almazan, A. and J. Suarez (2003) "Entrenchment and Severance Pay in Optimal Governance Structure", *Journal of Finance*, 58, 519-547.

Burkart, M., Gromb, D. and F. Panunzi (1997) "Large Shareholders, Monitoring and the Value of the Firm" *Quarterly Journal of Economics*, 112, 693-728.

Dominguez-Martinez, S., Swank, O. and B. Visser (2008) "In Defence of Boards" *Journal of Economics and Management Strategy*, 17, 667-682.

Faccio, M. and L. Lang ((2002) 'The Ultimate Ownership of Western European Corporation,' *Journal of Financial Economics*, 65, 365-395.

Graziano, C. and A. Luporini (2003) 'Board Efficiency and Internal Corporate Control Mechanisms', *Journal of Economics and Management Strategy*, 12, 495-530.

Harris, M. and A. Raviv (2008) 'A Theory of Board Control and Size,' *Review* of Financial Studies, 21, 1797-1832.

Hermalin, B.E. (2005) 'Trends in Corporate Governance', Journal of Finance, 60, 2351-2384.

Hermalin, B.E. and M. Weisbach (1998) 'Endogenously Chosen Boards of Directors and Their Monitoring of the CEO' American Economics Review, 88, 96-118.

Hermalin, B.E. and M. Weisbach (2003) 'Boards of Directors as an Endogenously Determined Institution: A Survey of the Economic Literature' *Economic Policy Review*, Vol. 9, 7–26.

Hirshleifer, D. and A. Thakor (1998) 'Corporate Control Through Dismissals and Takeovers', *Journal of Economics and Management Strategies*, 7, 489-520.

Holderness, C. (2009) " The Myth of Diffuse Ownership in the U.S." *Review* of Financial Studies, 22, 1377-1408.

Inderst, R. and H. Muller (2008) "CEO replacement under private information" Working Paper London School of Economics.

La Porta, R., Lopez-de-Silanes F., and A. Shleifer (1999) "Corporate Ownership around the world" *Journal of Finance*, 54, 471-517.

Manso G. (2008) "Motivating Innovation" Working Paper MIT.

Raheja, C.G. (2005) "Determinants of Board Size and Composition: a Theory of Corporate Boards", *Journal of Financial and Quantitative Analysis*, 40, 283-306.

Rusticus, T. (2006) "Executive Severance Agreements", Working paper University of Pennsylvania.

Shleifer, A. and R.W. Vishny (1997) 'A Survey of Corporate Governance' Journal of Finance, 52, 737-783.

Tirole, J. (2001) "Corporate Governance" Econometrica, 69, 1-35.

Warther, V.A. (1998) 'Board Effectiveness and Board Dissent: a Model of the Board's Relationship to Management and Shareholders', *Journal of Corporate Finance*, 4, 53-70.

Yermack, D. (2006) "Golden handshakes: separation pay for retired and dismissed CEOs" *Journal of Accounting and Economics*, 41, 237-256.

## 10 Appendix

#### 10.1 Proof of Proposition 1

Note that the maximization problem (9) can be written as:

$$\underset{\alpha,w,P^F}{Max} \qquad e_d A_d + \lambda \overline{\pi} + \alpha_d (1-\lambda)^2 \lambda^2 \overline{\pi}^2 - \frac{[\alpha_d (1-\lambda)\lambda \overline{\pi}]^2}{2} + e_d Z_d$$

s.t. (7), (8), (3), (1), (2) and where

$$A_d \equiv p_d [R - w \left(1 - F_i\right) - sF_d] \tag{17}$$

and

$$Z_{d} \equiv \lambda \overline{\pi} (1-\lambda)^{2} \alpha_{d} p_{d} (1-\lambda - P_{d}^{F}) (w_{d} - s_{d}) -$$

$$\frac{(1-\lambda)^{2} \alpha_{d}^{2} p_{d}^{2} (1-\lambda - P_{d}^{F})^{2} (w_{d} - s_{d})^{2}}{2} - \lambda \overline{\pi} (1-\lambda)^{2} \alpha_{d}^{2} p_{d} (1-\lambda - P_{d}^{F}) (w_{d} - s_{d}).$$
(18)

We ignore for the moment (1), (2). Once the solutions for w and s are derived, it will be clear that they are satisfied. The other constraints are substituted in the objective function.

The FOCs for w, s and  $\alpha$  are:

$$w: \qquad \frac{\partial e_d}{\partial w} (A_d + Z_d) + e_d \left[\frac{\partial A_d}{\partial w} + \frac{\partial Z_d}{\partial w}\right] = 0 \tag{19}$$

$$\frac{\partial e_d}{\partial s}(A_d + Z_d) + e_d \left[\frac{\partial A_d}{\partial s} + \frac{\partial Z_d}{\partial s}\right] = 0 \tag{20}$$

 $\alpha$  :

s :

$$\frac{\partial e_d}{\partial \alpha} (A_d + Z_d) + e_d \left[ \frac{\partial A_d}{\partial \alpha} + \frac{\partial Z_d}{\partial \alpha} \right] + \lambda^2 \overline{\pi}^2 (1 - \lambda)^2 (1 - \alpha) = 0 \quad (21)$$

where

$$\begin{split} \frac{\partial e_d}{\partial w} &= p_d \left(1 - F_d\right) - (w - s) \frac{\partial F_d}{\partial w} - db \frac{\partial F_1}{\partial w},\\ \frac{\partial e_d}{\partial s} &= p_d F_d - (w - s) \frac{\partial F_d}{\partial s} - db \frac{\partial F_1}{\partial s},\\ \frac{\partial e_d}{\partial \alpha} &= -p_d (w - s) \frac{\partial F_d}{\partial \alpha} - db \frac{\partial F_1}{\partial \alpha}, \end{split}$$

and

$$\frac{\partial F_d}{\partial w} = (1 - \lambda - P^F) \frac{\partial M_i}{\partial w} = -(1 - \lambda - P^F) \frac{\partial M_d}{\partial s} = -\frac{\partial F_d}{\partial s}$$
$$\frac{\partial F_d}{\partial \alpha} = (1 - \lambda - P^F) \frac{\partial M_d}{\partial \alpha}$$

where

$$\frac{\partial M_d}{\partial w} = \alpha (1 - \lambda) p_d (1 - \lambda - P^F) = -\frac{\partial M_d}{\partial s}.$$
$$\frac{\partial M_d}{\partial \alpha} = \left\{ (1 - \lambda) \left[ \lambda \overline{\pi} + I_{pro} p_d (1 - \lambda - P_d^F) (w_d - s_d) \right] \right\}$$

Summing up (19) and (20) and substituting for the derivatives from the above expressions we obtain:

$$A_d + Z_d = e_d \tag{22}$$

.

Substituting (22) into (19) and taking the derivatives we have:

$$db(1 - \lambda - P^F) = \lambda \overline{\pi}(1 - \lambda) - M_d \quad \text{with } I_{pro} = 1.$$
(23)

Considering that  $P^F$  must satisfy (3) and that  $\alpha$  must not exceed unity, i) for d = 0 condition (23) can be satisfied iff w = s and  $\alpha = 1$ , and ii) for d = 1 it cannot be satisfied if w > s.

Considering (17), (18), (7) and substituting w = s in (22) we obtain

$$w_0 = \frac{R}{2}$$

In order to fully characterize the case of d = 1, let us first substitute (22) into (21). Using (3), after some manipulation we have that, for  $w \leq s$ , (21) can then be written as:

$$(1-\alpha)\lambda^2 \overline{\pi}^2 (1-\lambda)^2 (1-e_1) = 0.$$

Considering that  $e_1 < 1$ , this implies  $\alpha = 1$ . Given  $\alpha = 1$ , (23) in turn implies:

$$s - w = \frac{b}{p_1(1 - \lambda)}$$
 and  $P_1^F = 0$ 

Considering (17), (18), (8) and substituting for s - w in (22) we obtain:

$$w_1 = \frac{R}{2} - \frac{b + (1 - \lambda)^2 \lambda \overline{\pi}}{2p_1} + \frac{b^2 (1 - \lambda)(1 + 3\lambda)}{4p_1}$$

Finally it can be easily verified that  $P_0^F \in [0, 1]$ .

#### 10.2 Proof of Proposition 3

 $e_0$  and  $e_1$  can be written as:

$$e_0 = p_0 \frac{R}{2}$$
$$e_1 = p_1 \frac{R}{2} + \frac{b}{2} [1 - (1 - \lambda)^2 (\lambda \overline{\pi} - b)] - \frac{(1 - \lambda)^2 b^2}{4}$$

which implies that  $e_0 > e_1$  when b = 0. Given that

$$\frac{\partial e_1}{\partial b} = \frac{1}{2} + (1-\lambda)^2 \frac{b}{2} > 0,$$

 $e_1$  however is increasing in b ranging from  $e_{1|b=0} = p_1 \frac{R}{2}$  when b = 0 to a level that tends to  $e_{1|b=\beta} = p_1 \frac{R}{2} + \frac{\lambda \overline{\pi}}{2} (1 - \lambda \overline{\pi} \frac{(1-\lambda)^2}{2})$  when  $b \to \beta \equiv (1-\lambda)\lambda \overline{\pi}$ . When  $b \to \beta$  we then have  $e_{1|b\to\beta} > e_0$  so that the first part of Proposition 3 is implied. Given that the difference between  $e_1$  and  $e_0$  is clearly increasing in  $(p_0 - p_1) \frac{R}{2}$ the second part of the proposition immediately follows.

#### 10.3 Proof of Proposition 4

The first part of the proposition immediately follows from  $e_{0|w=s=0} = 0$ . The values of  $\alpha_{1|w=s=0}$  and  $P_{1|w=s=0}^F$  are derived from the first-order conditions of the large shareholder's optimization problem that now take the form:

$$\alpha: \qquad -b(1-\lambda-P^F)(1-\lambda)\lambda\overline{\pi}p_1R + \lambda^2\overline{\pi}^2(1-\lambda)^2(1-\alpha) = 0$$

 $P^F$ :  $p_1 Rb(-1 + \alpha(1 - \lambda)\lambda\overline{\pi}) < 0$ 

implying  $P_{1|w=s=0}^F = 0$  and  $\alpha_{1|w=s=0} = 1 - \frac{bRp_1}{\lambda \overline{\pi}}$ .

#### 10.4 Proof of Proposition 5

The maximization problem of the large shareholder is the same as that analyzed in the proof of Proposition 1, with the only difference that for d = 1 managerial effort is given by:

$$\tilde{e}_1 = p_1 w (1 - F_1) + sF_1 + b$$

implying

$$\frac{\partial \tilde{e}_1}{\partial w} = p_1 \left( 1 - F_1 \right) - \left( w - s \right) \frac{\partial F_1}{\partial w}$$

which in turn implies

$$\frac{\partial \widetilde{e}_1}{\partial w} + \frac{\partial A_1}{\partial w} = 0$$

Then, if we proceed as in the proof of Proposition 1 and we substitute  $A_i + Z_i = e_i$  in (19) we now obtain:

$$\widetilde{e}_1 \frac{\partial Z_1}{\partial w} = 0$$

which implies

$$[\lambda \overline{\pi}(1-\alpha) - \alpha p_1(\widetilde{w}_1 - \widetilde{s}_1)(1-\lambda - P^F)] = 0.$$

Considering that  $P^F$  must satisfy (3) and that  $\alpha$  must not exceed unity, this expression implies

$$\widetilde{w}_1 - \widetilde{s}_1 = 0$$
 and  $\alpha = 1$ .

Given  $\widetilde{w}_1 = \widetilde{s}_1$  any value of  $P_1^F \in [0, 1]$  satisfies the FOCs. Using  $A + Z = \widetilde{e}_1$ , which simplifies to  $A = \widetilde{e}_1$  when w = s, we obtain:

$$\widetilde{w}_1 = \widetilde{s}_1 = \frac{R}{2} - \frac{b}{2p_1}$$

from (19).

#### 10.5 Proof of Proposition 7

This proof follows the same line of reasoning used in the proof of Proposition 3. Also in this case the equity value is larger under delegation if and only if managerial effort is larger: i.e.,  $\tilde{V}_1 > \tilde{V}_0$  iff  $\tilde{e}_1 > \tilde{e}_0 = e_0$ . Given that  $e_0 = \frac{Rp_0}{2}$ 

and  $\tilde{e}_1 = \frac{Rp_1}{2} + \frac{b}{2}$  clearly  $\tilde{e}_1 < \tilde{e}_0$  when b = 0. It is immediate to see that  $\tilde{e}_1$  is increasing in b. When  $b \to \beta$ ,  $\tilde{e}_1 > \tilde{e}_0$  because  $\beta > R(p_0 - p_1)$ . Given that the difference between  $\tilde{e}_1$  and  $\tilde{e}_0$  is clearly increasing in  $(p_0 - p_1) \frac{R}{2}$ , the second part of the proposition immediately follows.  $\tilde{b} < \bar{b}$  follows from  $\tilde{e}_1 > e_1$  when b > 0.

#### 10.6 Proof of Proposition 8

The results for d = 1 are those derived in Proposition 5. When d = 0 we can replicate the same procedure used in the proof of Proposition 1 considering that now in the expression for A we also have the constant B:

$$\widehat{A} \equiv \{ p_0 [R - w_0 (1 - F_0) - s_0 F_0] + B \}.$$

As a consequence, after having verified that w = s, we set  $\widehat{A} = \widehat{e}_0$  and we obtain:

$$\widehat{w}_0 = \widehat{s}_0 = \frac{R}{2} + \frac{B}{2p_0}.$$

#### 10.7 Proof of Proposition 9

This proof follows the same line of reasoning used in the Proof of Proposition 3 and 7. Again, the equity value is larger under delegation if and only if managerial effort is larger: i.e.  $\hat{V}_1 > \hat{V}_0$  iff  $\hat{e}_1 > \hat{e}_0$ . Given the values of  $\hat{e}_1$  and  $\hat{e}_0$ , this implies

$$\frac{Rp_1}{2} + \frac{b}{2} - \frac{Rp_0}{2} - \frac{B}{2} > 0$$

or

$$b - B > (p_0 - p_1) R.$$

## **CESifo Working Paper Series**

for full list see www.cesifo-group.org/wp (address: Poschingerstr. 5, 81679 Munich, Germany, office@cesifo.de)

- 2961 Oliver Falck, Stephan Heblich, Alfred Lameli and Jens Suedekum, Dialects, Cultural Identity, and Economic Exchange, February 2010
- 2962 Bård Harstad, The Dynamics of Climate Agreements, February 2010
- 2963 Frederick van der Ploeg and Cees Withagen, Is There Really a Green Paradox?, February 2010
- 2964 Ingo Vogelsang, Incentive Regulation, Investments and Technological Change, February 2010
- 2965 Jan C. van Ours and Lenny Stoeldraijer, Age, Wage and Productivity, February 2010
- 2966 Michael Hoel, Climate Change and Carbon Tax Expectations, February 2010
- 2967 Tommaso Nannicini and Roberto Ricciuti, Autocratic Transitions and Growth, February 2010
- 2968 Sebastian Brauer and Frank Westermann, A Note on the Time Series Measure of Conservatism, February 2010
- 2969 Wolfram F. Richter, Efficient Education Policy A Second-Order Elasticity Rule, February 2010
- 2970 Tomer Blumkin, Yoram Margalioth and Efraim Sadka, Taxing Children: The Redistributive Role of Child Benefits – Revisited, February 2010
- 2971 Chang Woon Nam and Georg Wamser, Application of Regionally Varying Additionality Degrees in the Practice of EU Cohesion Policy, February 2010
- 2972 Ali Bayar, Frédéric Dramais, Cristina Mohora, Masudi Opese and Bram Smeets, Modeling Russia for Climate Change Issues, February 2010
- 2973 Magnus Söderberg, Informal Benchmarks as a Source of Regulatory Threat in Unregulated Utility Sectors, March 2010
- 2974 Piotr Wdowiński and Marta Malecka, Asymmetry in Volatility: A Comparison of Developed and Transition Stock Markets, March 2010
- 2975 Frans van Winden, Michal Krawczyk and Astrid Hopfensitz, Investment, Resolution of Risk, and the Role of Affect, March 2010
- 2976 Hyun-Ju Koh and Nadine Riedel, Do Governments Tax Agglomeration Rents?, March 2010

- 2977 Johann K. Brunner and Susanne Pech, Optimum Taxation of Bequests in a Model with Initial Wealth, March 2010
- 2978 Guglielmo Maria Caporale and Nicola Spagnolo, Stock Market Integration between three CEECs, Russia and the UK, March 2010
- 2979 Florian Englmaier, Ales Filipi and Ravi Singh, Incentives, Reputation and the Allocation of Authority, March 2010
- 2980 Konstantinos Angelopoulos, George Economides and Apostolis Philippopoulos, What is the Best Environmental Policy? Taxes, Permits and Rules under Economic and Environmental Uncertainty, March 2010
- 2981 Frederick van der Ploeg, Rapacious Resource Depletion, Excessive Investment and Insecure Property Rights, March 2010
- 2982 Wolfram F. Richter and Christoph Braun, Efficient Subsidization of Human Capital Accumulation with Overlapping Generations and Endogenous Growth, March 2010
- 2983 Francesco Cinnirella, Marc Piopiunik and Joachim Winter, Why Does Height Matter for Educational Attainment? Evidence from German Pre-Teen Children, March 2010
- 2984 Bernard Van Praag, Well-being Inequality and Reference Groups An Agenda for New Research, March 2010
- 2985 Francesca Barion, Raffaele Miniaci, Paolo M. Panteghini and Maria Laura Parisi, Profit Shifting by Debt Financing in Europe, March 2010
- 2986 Alexander Haupt and Magdalena Stadejek, The Choice of Environmental Policy Instruments: Energy Efficiency and Redistribution, March 2010
- 2987 John Komlos and Marek Brabec, The Trend of BMI Values among US Adults, March 2010
- 2988 Emanuele Massetti and Lea Nicita, The Optimal Climate Policy Portfolio when Knowledge Spills across Sectors, March 2010
- 2989 Helmut Rainer and Thomas Siedler, Family Location and Caregiving Patterns from an International Perspective, March 2010
- 2990 Toru Kikuchi and Ngo Van Long, A Simple Model of Service Offshoring with Time Zone Differences, March 2010
- 2991 Assaf Razin, Efraim Sadka and Benjarong Suwankiri, Migration and the Welfare State: Dynamic Political-Economy Theory, March 2010
- 2992 Bård Harstad, Buy Coal! Deposit Markets Prevent Carbon Leakage, March 2010
- 2993 Axel Dreher, Stephan Klasen, James Raymond Vreeland and Eric Werker, The Costs of Favoritism: Is Politically-driven Aid less Effective?, March 2010

- 2994 Sven Neelsen and Thomas Stratmann, Effects of Prenatal and Early Life Malnutrition: Evidence from the Greek Famine, March 2010
- 2995 Claude Hillinger and Bernd Süssmuth, The Quantity Theory of Money: An Assessment of its Real Linchpin Prediction, March 2010
- 2996 Matthew M. Chingos and Martin R. West, Do More Effective Teachers Earn More Outside of the Classroom?, March 2010
- 2997 Laurence Jacquet and Dirk Van de gaer, A Comparison of Optimal Tax Policies when Compensation or Responsibility Matter, March 2010
- 2998 Valentina Bosetti, Carlo Carraro, Romain Duval and Massimo Tavoni, What Should we Expect from Innovation? A Model-Based Assessment of the Environmental and Mitigation Cost Implications of Climate-Related R&D, March 2010
- 2999 Scott Alan Carson, Nineteenth Century Stature and Family Size: Binding Constraint or Productive Labor Force?, March 2010
- 3000 Jukka Pirttilä and Ilpo Suoniemi, Public Provision, Commodity Demand and Hours of Work: An Empirical Analysis, March 2010
- 3001 Bertrand Candelon and Franz C. Palm, Banking and Debt Crises in Europe: The Dangerous Liaisons?, March 2010
- 3002 Joan Costa-i-Font and Marin Gemmill-Toyama, Does Cost Sharing really Reduce Inappropriate Prescriptions?, March 2010
- 3003 Scott Barrett, Climate Treaties and Backstop Technologies, March 2010
- 3004 Hans Jarle Kind, Tore Nilssen and Lars Sørgard, Price Coordination in Two-Sided Markets: Competition in the TV Industry, March 2010
- 3005 Jay Pil Choi and Heiko Gerlach, Global Cartels, Leniency Programs and International Antitrust Cooperation, March 2010
- 3006 Aneta Hryckiewicz and Oskar Kowalewski, Why do Foreign Banks Withdraw from other Countries? A Panel Data Analysis, March 2010
- 3007 Eric A. Hanushek and Ludger Woessmann, Sample Selectivity and the Validity of International Student Achievement Tests in Economic Research, March 2010
- 3008 Dennis Novy, International Trade and Monopolistic Competition without CES: Estimating Translog Gravity, April 2010
- 3009 Yin-Wong Cheung, Guonan Ma and Robert N. McCauley, Renminbising China's Foreign Assets, April 2010
- 3010 Michel Beine and Sara Salomone, Migration and Networks: Does Education Matter more than Gender?, April 2010

- 3011 Friedrich Schneider, Tilman Brück and Daniel Meierrieks, The Economics of Terrorism and Counter-Terrorism: A Survey (Part I), April 2010
- 3012 Friedrich Schneider, Tilman Brück and Daniel Meierrieks, The Economics of Terrorism and Counter-Terrorism: A Survey (Part II), April 2010
- 3013 Frederick van der Ploeg and Steven Poelhekke, The Pungent Smell of "Red Herrings": Subsoil Assets, Rents, Volatility and the Resource Curse, April 2010
- 3014 Vjollca Sadiraj, Jan Tuinstra and Frans van Winden, Identification of Voters with Interest Groups Improves the Electoral Chances of the Challenger, April 2010
- 3015 Guglielmo Maria Caporale, Davide Ciferri and Alessandro Girardi, Time-Varying Spot and Futures Oil Price Dynamics, April 2010
- 3016 Scott Alan Carson, Racial Differences in Body-Mass Indices for Men Imprisoned in 19<sup>th</sup> Century US Prisons: A Multinomial Approach, April 2010
- 3017 Alessandro Fedele, Paolo M. Panteghini and Sergio Vergalli, Optimal Investment and Financial Strategies under Tax Rate Uncertainty, April 2010
- 3018 Laurence Jacquet, Take it or Leave it: Take-up, Optimal Transfer Programs, and Monitoring, April 2010
- 3019 Wilhelm Kohler and Jens Wrona, Offshoring Tasks, yet Creating Jobs?, April 2010
- 3020 Paul De Grauwe, Top-Down versus Bottom-Up Macroeconomics, April 2010
- 3021 Karl Ove Aarbu, Demand Patterns for Treatment Insurance in Norway, April 2010
- 3022 Toke S. Aidt and Jayasri Dutta, Fiscal Federalism and Electoral Accountability, April 2010
- 3023 Bahram Pesaran and M. Hashem Pesaran, Conditional Volatility and Correlations of Weekly Returns and the VaR Analysis of 2008 Stock Market Crash, April 2010
- 3024 Stefan Buehler and Dennis L. Gärtner, Making Sense of Non-Binding Retail-Price Recommendations, April 2010
- 3025 Leonid V. Azarnert, Immigration, Fertility, and Human Capital: A Model of Economic Decline of the West, April 2010
- 3026 Christian Bayer and Klaus Wälde, Matching and Saving in Continuous Time: Theory and 3026-A Matching and Saving in Continuous Time: Proofs, April 2010
- 3027 Coen N. Teulings and Nick Zubanov, Is Economic Recovery a Myth? Robust Estimation of Impulse Responses, April 2010
- 3028 Clara Graziano and Annalisa Luporini, Optimal Delegation when the Large Shareholder has Multiple Tasks, April 2010